Author's response to reviews

Title: Association between 24-hour blood pressure variability and chronic kidney disease: A cross-sectional analysis of African Americans participating in the Jackson Heart Study

Authors:

Rikki M Tanner (rmdeitz@uab.edu)
Daichi Shimbo (ds2231@cumc.columbia.edu)
Albert W Dreisbach (adreisbach@umc.edu)
April P Carson (apcarson@uab.edu)
Ervin R Fox (efox@umc.edu)
Paul Muntner (pmuntner@uab.edu)

Version: 2
Date: 20 February 2015

Author's response to reviews:

Thank you for your preliminary review of our manuscript. We have revised the manuscript to include line numbers and an abbreviations section. We have also made a separate "conclusions" section, per your instructions.